Author's response to reviews

Title: Strategies to prevent intraoperative lung injury during cardiopulmonary bypass

Authors:

Efstratios Apostolakis (stratisapostolakis@yahoo.gr)
Efstratios Koletsis (ekoletsis@hotmail.com)
Nikolaos G Baikoussis (ngbaik@yahoo.com)
Stavros Siminelakis (ngbaik@yahoo.com)
Georgios Papadopoulos (ngbaik@yahoo.com)

Version: 2 Date: 6 December 2009

Author's response to reviews:

Patras 6/12/09

Editor in Chief
Journal of cardiothoracic surgery

Cover letter

Dear Editor

We herewith submit the corrections of our review article according the reviewers comments. We changed the title as the recommendation, and we corrected all grammars error indicated by reviewers of the Journal of cardiothoracic surgery. A shortening of the manuscript was also done.

The paper is prepared following the instructions of Journal of cardiothoracic surgery on line.

Sincerely yours

N.G.Baikoussis MD